Contents lists available at ScienceDirect



Environmental Innovation and Societal Transitions

journal homepage: www.elsevier.com/locate/eist

Heating in Great Britain: An incumbent discourse coalition resists an electrifying future



Richard Lowes^{a,*}, Bridget Woodman^b, Jamie Speirs^c

^a Fellows Office, Stella Turk Building, University of Exeter, Penryn Campus, Penryn, TR10 9FE, UK
 ^b B051-033, Stella Turk Building, University of Exeter, Penryn Campus, Penryn, TR10 9FE UK
 ^c 537, Chemistry, South Kensington Campus, Imperial College London, SW7 2AZ, UK

ARTICLE INFO

Keywords: Power Incumbency Discourse Coalition Heat Gas

ABSTRACT

The power of incumbent actors to affect sustainability transitions is increasingly recognised as a central issue associated with systemic change. However, incumbent's approaches and the outcome of their influence is rarely examined in academic literature. Using a novel approach which combines the lens of 'discourse coalitions' with an explicitly critical discursive stance, in which the coalition's storyline is scrutinised, this interdisciplinary analysis investigates a pro-gas, in cumbent led coalition present in the Great Britain (GB) energy system. In response to the threat of electrification, the coalition presents decarbonising the gas grid with replacement gases as the optimal route for heat decarbonisation. However, much analysis suggests a significant need for heat electrification and our review highlights major uncertainties with a decarbonised gas pathway. Incumbents are over-selling 'green-gas' to policy makers in order to protect their interests and detract from the importance and value of electrification. Policy and research recommendations are made.

1. Introduction

The transitions of systems from unsustainability to sustainability are likely to challenge existing regime interests who may resist change (Geels, 2014). The required transformation¹ of the global energy system to zero carbon, in line with the Paris Agreement, appears particularly radical in both speed and scope (IPCC, 2018). The most radical transformations may be particularly at risk of attempts at subversion and 'aims in social science' should act 'strongly to resist the shaping of knowledge by incumbent interests' (Stirling, 2014, p91).

This article investigates attempts by incumbent interests to shape GB heat decarbonisation policy and questions the validity of their proposed storyline. We use a critical discursive approach, and borrow the lens of 'discourse coalitions' (DCs) in order to take a highly policy relevant analytical methodology which takes into account the state of the art analysis around UK heat decarbonisation

* Corresponding author.

E-mail addresses: r.lowes@exeter.ac.uk (R. Lowes), b.woodman@exeter.ac.uk (B. Woodman), jamie.speirs@imperial.ac.uk (J. Speirs).

¹ While much of the literature on sustainable change uses the term 'transition', when considering the radical changes required for the global and Great Britain heat system we use to term 'transformation' to emphasize the speed and breadth of the required changes.

https://doi.org/10.1016/j.eist.2020.07.007

Received 8 January 2020; Received in revised form 24 July 2020; Accepted 27 July 2020

2210-4224/ © 2020 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/BY/4.0/).

Abbreviations: AD, Anaerobic digestion; ADA, Argumentative discourse analysis; BEIS, Department for Business, Energy and Industrial Strategy; CCC, Committee on Climate Change; CCS, Carbon capture and storage; CDA, Critical discourse analysis; DC, Discourse coalition; DECC, Department of Energy and Climate Change; ENA, Energy Networks Association; GHG, Greenhouse gas; GB, Great Britain; LPG, Liquefied petroleum gas; MLP, Multi-level perspective; NGO, Non-governmental organisation; SNG, Synthetic natural gas; UK, United Kingdom

pathways.

While this study focuses on GB, heat decarbonisation is a global challenge and many socio-technical issues associated with GB's heat transition may be reflected in other countries, in particular those with large penetrations of fossil gas heating. This article may therefore have particular value for scholars interested in the heat transitions in the Netherlands, Italy and certain US states. In California, popular media is already reporting corporate resistance against heat electrification (LA Times, 2019) and at a US-wide level, recent analysis by environmental NGOs has highlighted the promotion of 'Renewable Natural Gas' in response to ideas of electrification (Earth Justice and The Sierra Club, 2020).

1.1. Heating in the United Kingdom (UK) and its future

Greenhouse gas (GHG) emissions from space heating and hot water production in the UK^2 make up a fifth of emissions (Committee on Climate Change, 2016) with 79 % of this energy provided by the combustion of fossil fuels (BEIS, 2018a). Net zero GHG emission targets imply total elimination of these emissions requiring the replacement of all fossil fuel heating systems with something else alongside reductions in heat demand (Committee on Climate Change, 2019a).

Techno-economic analysis by academia, the UK Government and its advisors has repeatedly suggested that electrification of much heat demand, primarily using heat pumps alongside the reduction in heat demand, represents the lowest cost pathway to near fully decarbonised heating (Committee on Climate Change, 2010; DECC, 2013, 2012, 2010; Element Energy and AEA, 2012; UKERC, 2009). Despite some uncertainty, there are common messages for the future of UK heating:

- 'Energy demand reduction is essential for meeting emission targets'
- 'A substantial level of electrification of heating (via heat pumps) is expected'
- 'District heating will play an important role in heat supply decarbonisation'

(Chaudry et al., 2015, p268).

The UK's statutory climate advisor 'The Committee on Climate Change' (CCC) sees 19 million electrically powered heat pumps (including air source, ground source and hybrids) installed in UK homes in 2050 in its 'further ambition' scenario which aims towards net zero emissions; this growth in heat pump systems is expected to come alongside growth in heat networks, reductions in heat demand through energy efficiency measures (Committee on Climate Change, 2019a). Recent analysis has shown that heat pumps already out-compete fossil fuel heating based on carbon intensity in the majority of situations globally (Knobloch et al., 2020) and that the potential for decarbonisation of heat using heat pumps across Europe would provide significant and immediate carbon benefits (Rosenow and Lowes, 2020).

The heat electrification transformation would have significant impacts on incumbents in the UK's heat sector who may no longer be able to sell gas boilers or may own stranded assets such as gas infrastructure (Lowes et al., 2018a). Some response by incumbents therefore seems likely.

1.2. Increasing complexity

Despite the expected importance of electrification, converting the UK's gas system to run on low carbon gases including hydrogen and biomethane (often referred to as 'green gases') has recently and rapidly emerged as a key technology option considered by policy makers to decarbonise heating in the UK (Committee on Climate Change, 2018a; HM Government, 2017). This gas-based pathway has emerged in the UK heat policy discourse at the same time as strong promotion of this technological approach by the gas industry suggesting the gas industry may have had some political success.

However, it may also be that genuine concerns over the required increase in electricity system capacity which would result from heat electrification (which have not just been highlighted by incumbents e.g. Maclean et al., 2016 have led to his complexity. In general, heat decarbonisation pathways in the UK have become more complex with pathways which now include full electrification, full conversion to hydrogen, hybrid pathways and pathways with geographical variation (Lowes et al., 2018a; Winskel et al., 2016).

Electrifying the heat currently provided by gas would have an impact on the electricity system with analysis suggesting that if 20 % of homes had heat pumps, electricity system peak demand could increase by 14 % (Love et al., 2017). However, while electrifying heat would clearly increase electricity system throughput, recent analysis suggests these 'peak heat' concerns may be less of an issue than was previously believed (Watson et al., 2019). Others have also added that reductions in heat demand alongside 'smart' heat load shifting approaches can reduce system impacts further (Rosenow and Lowes, 2020). It may of course also be that increasing GB electricity capacity is simply a requirement of heat decarbonisation.

Recent analysis suggested the most cost effective GB heat pathway to net-zero may be one where much heat demand is electrified but this is supported by hybrid systems which top up heat pumps with gas³ based heating (Strbac et al., 2018). These hybrids work at

 $^{^{2}}$ The UK (The United Kingdom of Great Britain and Northern Ireland) includes Northern Ireland and the Isle of Man. Energy governance for the UK and GB is connected and while we discuss the UK throughout this paper where relevant, for example in relation to analysis or system change, the empirical focus of this work is GB.

 $^{^{3}}$ This would need to be some form of low carbon gas with hydrogen seen as the vector. This hybrid model would therefore require the deployment of heat pumps at scale and the conversion of the gas distribution networks to hydrogen alongside the use of boilers suitable for hydrogen.

a building level requiring the maintenance of much gas infrastructure and the need for two appliances. It has also been suggested that because costs between technological pathways may be similar, low carbon heat solutions may vary by location depending on, for example, available resources and infrastructure (Committee on Climate Change, 2018a). Nonetheless, because of limited deployment and a lack of practical experience, significant uncertainty remains over the potential of low carbon gas.

In light of the apparent uncertainty over the optimal pathway, the UK government is not expecting to make strategic decisions on the future of heating, particularly for houses on the gas grid, until the evidence around options, electrification or hydrogen, provides greater clarity on the optimum solution (HM Government, 2017). Meanwhile, the deployment of energy efficiency and heat pumps in the UK is below indicator levels set by the CCC for decarbonisation (Committee on Climate Change, 2019b). Recent analysis has also shown that heat policy makers in GB perceive technological approaches for heat decarbonisation as both disruptive, but particularly important for this analysis, uncertain (Lowes and Woodman, 2020).

The lack of progress around heating and uncertainty over future technology suggests goals for decarbonisation may be at risk. This article is interested in the potential role of incumbents in causing this risk.

1.3. Research focus

This article considers the behaviour of incumbents⁴ threatened by heat decarbonisation, through the examination of the green gas discourse, which they are promoting.

Fundamentally, this article is interested in the shape and practices of the incumbent coalition as well as the discursive storyline that is being promoted. Through a critical discursive approach, this interdisciplinary investigation considers the 'green gas' storyline and its proponents alongside technical analysis associated with so-called 'green gas' and heat decarbonisation pathways.

Specifically, we investigate:

- 1 Can a DC be identified and what practices are being used to promote the storyline?
- 2 How does the proposed storyline compare to relevant and recent analysis on UK heat decarbonisation pathways and the potential role for gas?
- 3 What are the transition and policy implications of the efforts of incumbents?

Section 2 explores the theory of incumbency, discourse coalitions and critical discourse analysis. Section 3 explains research materials and methods. Section 4 considers the presentation of the GB 'green gas' DC. Based on a state of-the-art evidence review, section 5 critiques the 'green gas' storyline employed by the discourse coalition. Section 6 considers the potential impacts of the approaches of the incumbent discourse coalition on GB heat decarbonisation. Section 7 concludes the paper and highlights the implications for policy.

2. Power, incumbency and critical discourse approaches

The power of actors to affect the transitions of large socio-technical systems, such as the UK heat transformation, is widely regarded as significant (Avelino and Rotmans, 2011, 2009; Geels, 2014, 2010; Kern, 2011; Smith et al., 2005) and the literature investigating issues of power and sustainable change is rapidly growing. However while the concept of power is a 'core concept' in the social sciences, it is 'arguably one of the most difficult concepts to make sense of' (Clegg and Haugaard, 2009, p1).

A lack of a definitive characterisation of power and the complexity of the issue means that it 'seems as if any student of power designs his or her approach' (Arts, 2000, p112). This article takes an 'interests' based approach whereby actors are expected to behave in a way which maximises 'personal utility' (Kern, 2011, p1120) and for the purpose of this article, power can be considered as the ability of an actor to affect the transition to sustainable heating. If actors are able to have power to affect the UK's heat transformation, this could have implications for the speed and direction of change.

Smith et al. (2010) suggest that 'opening the black-box of public policy' may have particular value for scholars interested in how power can affect transitions because the policy process may be a particuarly important location of power struggles. They go on to explain that 'there are long-standing literatures on regulatory capture, government-industry relations, clientelism, iron triangles, policy networks, and discourse coalitions that can help us ensure analysis of socio-technical regimes and public policies are more deeply intertwined' (p446). This paper makes a contribution to opening the policy black-box through its critique on the practices of a specific discourse coalition.

The relationships between system transitions and power have been considered previously. Investigations have considered the power of discourses and ideas (Castán Broto, 2015; Späth and Rohracher, 2010), the power associated with historical regimes (Arapostathis et al., 2013), the power associated with institutions (Fuenfschilling and Truffer, 2014) and a whole special issue of this journal has been dedicated to the 'politics of innovation spaces for low-carbon energy' (Raven et al., 2016). Lowes et al. (2019) specifically considered the power of actors to affect historic GB heat transformation policy development.

⁴ Lowes et al., (2017) develop a working definition of the term incumbency on which the view of incumbency in this paper is based. They '...define incumbency in the context of sustainable transformations as the presence of existing actors within a specific socio-technical system. An incumbent will be currently active in the socio-technical system or a part thereof and therefore likely to be or have been involved in unsustainable practices. Incumbents have the economic, social or technological capacity to influence system change'.

2.1. Incumbency

Markard (2018) suggested that the politics of transitions and specifically issues with incumbents attempting to slow down transitions will be a key area for energy transition research and has called for the adoption of political science concepts and approaches by transition scholars. Ideas of 'incumbency' and 'incumbents' have featured widely within the literature on power and sustainability transitions and the key issue for analysis of incumbents is whether incumbents have power to drive, slow or shape transitions.

Incumbency is widely seen as an issue which can slow or inhibit change (Johnstone et al., 2017; Roberts et al., 2018; Stirling, 2014; Unruh, 2000). This may however not always be the case with evidence existing of incumbents driving sustainability practices (Geels, 2011; Stenzel and Frenzel, 2008) and incumbents from neighbouring systems able to support change beyond their own regime (Turnheim and Geels, 2019).

Incumbent responses are also likely to depend on relative landscape pressures (Roberts et al., 2018), after all, why would an incumbent with a product or service that supports (normative) change attempt to block a transition? Indeed, some have suggested that the urgency of decarbonisation means that where incumbent regimes can support transitions, supportive management may be warranted (Winskel and Radcliffe, 2014). Overall, while incumbents appear to be important and powerful, their responses may vary.

Evidence on the actual impacts of the power of incumbent actors in response to transitions is limited. Smith et al. (2005) recognised the ability of regime actors to shape visions and expectations of transformations. Dutch transitions management approaches to energy system change have seen a dominance of so called transition arenas by incumbent interests (Hendriks and Grin, 2007; Kemp et al., 2007; Kemp and Smith, 2008). Recently, Lockwood et al. (2020) have shown that the development of the UK's capacity market for electricity appears to have been influenced, by, and for the benefit of, incumbent electricity generation interests and have proposed more independent and transparent regulation of this sector. There appears to be little, if any, research into incumbency associated with sustainable heat transformations, a gap we fill.

2.2. Critical discursive approaches

Stirling (2014) explicitly states that energy social science research must 'strongly resist the shaping of knowledge by incumbent interests' (p91).

Future visions of socio-technical systems may be an important element of transitions providing some sort of guide or story for actors to coalesce around (Avelino, 2009; Kenis et al., 2016; Turnheim and Geels, 2019). However, power relations and existing structures can limit or alter visions, potentially leading to long term transition impacts (Smith et al., 2005; Stirling, 2014). If incumbents can capture and therefore shape visions, impacts could be significant. Scholars have recently highlighted the use of 'discourses of delay' to reduce support for climate policies (Lamb et al., 2020).

Discursive approaches focus on how the social world is constructed through social elements including language, texts and the media (Burnham et al., 2008). So-called 'critical discourse analysis' (CDA) looks in part to understand how discourse relates to and potentially maintains inequality in power relations (Van Dijk, 1993a, 1993b). CDA may therefore have value for the consideration of large, systemically embedded incumbents (such as those within socio-technical (energy) systems) and how they are shaping visions and debates.

Even as a subset of wider approaches to discourse, CDA is a large research field however some general characteristics of CDA have been suggested:

- 1 It is not just analysis of discourse but is part of a more systematic transdisciplinary analysis of relations between discourse and other social elements.
- 2 It is not just a commentary on discourse but it includes some from of discursive analysis.
- 3 It is not just descriptive but it is also normative, addressing social wrongs in discursive form and suggesting possible ways to approach those wrongs.

(Fairclough, 2010)

Others agree that CDA is a broad approach but one which is 'problem oriented' while being focused on dominace and inequality in power relations (Van Dijk, 1993a, 1993b; Van Dijk, 1995). Van Dijk (2001) explains: 'CDA focuses on the ways discourse structures enact, confirm, legitimate, reproduce, or challenge relations of power and dominance in society' (p353) and also highlights the idea of using power to 'control the acts and minds of (members of) other groups' (p355). Taking this even further, Van Dijk (1993a, 1993b) suggests that CDA is a truly action oriented endeavour: 'Ultimately, its success is measured by its effectiveness and relevance, that is, by its contribution to change' (p253).

Methodologically, CDA approaches are diverse and foci can range from education to racism and also include political discourse and ideology (Blommaert and Bulcaen, 2000). In the case of public policy, these approaces focus on how discourse is shaped by social practice (Hewitt, 2009) and how discourses may dominate and show stability or instability (Fairclough, 2013). While not generally the key cause of policy changes, policy discourses are recognised as being be an influential factor (Hajer, 1995). With both limited progress, a lack of significant policy and governance support for low carbon heat and the perception of various and competing options for heat decarbonsation technologies in the UK, a CDA approach may have great value.

2.3. Discourse coalitions

CDA appears likely to have significant value for the study of incumbency associated with energy system change yet beyond the use of textual and interview analysis, there are no firm methodlogical approaches. In order to support this analysis with a relevant discursive lens, we borrow an approach based around the idea of discourse coalitions (DC).

DC approaches are often associated with argumentative discourse analytical (ADA) approaches whereby various DCs are considered to form a discursive space (Leifeld and Haunss, 2012). Discursive struggles take place with DCs competing to reach some form of hegemony or ideological dominance (Hajer, 2002). ADA approaches do not typically focus on a single discourse coalition or group of actors (such as incumbents) but there may be merit in combining the 'discourse coalitions' element of ADA with critical discursive approaches. This combined approach can provide focus on a particular group of organised actors while simultaneously unpicking the story that group is promoting and may have particular value for those investigating incumbent resistance.

According to Hajer (1995, p146):

'Discourse should be understood as a specific ensemble of ideas, concepts and categorizations that is produced, reproduced and transformed in a particular set of practices and through which meaning is given to physical and social realities. Discourse coalitions' refers to the way in which a particular discourse gets its social power. Here the term 'coalition' is meant to underline that this is not necessarily a matter of concerted and strategically negotiated action but might be the result of far more pragmatic, incidental alliances that shape up around specific 'story-lines''.

Hajer (1993) links the DC approach to issues associated with the mobilisation of bias and the so-called 'second face of power', an element of power associated with the ability of actors to set political agendas⁵.

Storylines have elsewhere been described as the '*intentional mobilisation of resource*' as a means of influencing policy change (Smith and Kern, 2009, p2001). Bulkeley (2000) suggests that in the case of DCs, despite consistent storylines, members may not necessarily share similar world views but the actors '*coalesce around certain storylines in order to advance oppositon to the dominant coaliton*' (p745). Lovell et al. (2009) agree, suggesting that storylines may be used by whoever they have value for.

DC approaches have frequently been used to consider large system dynamics. This incudes around wind energy policy (Szarka, 2004) and developments (Jessup, 2010), sustaianble housing (Lovell, 2008), the emergence of 'transition' ideas in Dutch policy making (Smith and Kern, 2009), shale gas (Cotton et al., 2014) and more recently around ideas of responsibility for food waste (Welch et al., 2018).

2.4. Theoretical synthesis

CDA and DC approaches appear to have significant value for investigating and framing the behaviours of incumbents in response to systemic transitions. CDA approaches encourage normatively critical approaches to discourse and DC approaches provide a lens to consider how actors can coalesce to attempt to dominate (Bulkeley, 2000).

A recent review highlighted the increase in the use of discursive approaches to consider energy transitions; this review showed the frequent use of argumentative discourse analysis but also showed the use of CDA (Isoaho and Karhunmaa, 2019). Analysis of Isoaho and Karhunmaa's (2019) data set shows five⁶ energy transition articles employing some form of 'critical' discursive approach. All of these articles which had a clear methodology used some form of text and/or interview based analysis.

These articles highlighted some key risks associated with policy discourses including potential overly-generous political support for bio-energy (Chaliganti and Müller, 2016), faith in technological developments to negate sustainability issues with bio-energy (Levidow and Papaioannou, 2013), politically challenging EU and USA discourses around the Chinese solar energy strategy (Caprotti, 2015) and competing views on geoengineering (Anshelm and Hansson, 2014).

In these articles, actual critique of the discourses under study appears often only superficial. Chaliganti and Müller (2016) went some way to highlight the way institutionalised discourses may crowd out other valuable approaches, echoing ideas of the importance of combining CDA methods with (new) institutional approaches as suggested by Genus (2014). Levidow and Papaioannou (2013) suggested real concerns over sustainability may be unheeded because of discursive ideas of 'future innovation' (p47). Overall however, only slight attention appears to have been paid to critique of the discourses under investigation.

With regards to DC approaches, Späth and Rohracher (2010) suggest that this approach has already been fruitfully linked with the 'multi-level perspective (MLP) on transition processes' (p252). However, the examples proided by Späth and Rohracher (2010) show in some cases only limited connections between the two approaches; for example, Smith et al. (2005) do not actually used the term 'discourse coalition' and Smith and Kern (2009) consider the DC associated with Dutch transitions approaches rather than considering DCs in approaches to sustainability transitions in general. And while Lovell et al. (2009) do consider ideas of transitions, they do not refer to the so called 'multi-level perspective'. So while there has been some cross-over between the approaches of transitions and DCs, this cross-over actually appears quite limited.

In order to build on this existing literature, and overcome some of the limits of previous research, a combination of CDA and DC

⁵ The 'faces of power' approach is one approach to consider power. Lukes describes 3 faces (Lukes, 2005) although others have suggested there are four faces to power, essentially the three faces of Lukes plus post-structural approaches to power (which include Foucauldian approaches) (Haugaard, 2012).

⁶ The original research article suggested seven articles were based on CDA but on review, not all were. Discounted articles include Lassen (2016) who focuses on 'genre' and (Schmid, 2004) who very briefly mentions CDA.

approaches may provide a useful approach to critically investigate the behaviour of incumbents in response to threat of energy system change.

Our approaches uses CDA as a basis but augments this with DC approaches where we specifically consider the make-up of the coalition and the story lines being promoted. The critical element of our analysis combines the storyline(s) being promoted with state of the art analysis around heat system decarbonisation and 'low carbon gas'. This multi-disciplinary analysis has been enabled by collaborative working across institutions in order to build on expertise around both incumbency and technical challenges associated with heat decarbonisation.

3. Materials and methods

All apart from one of the articles into DCs discussed in the previous section used a combination of interviews and grey literature analysis (Bulkeley, 2000; Cotton et al., 2014; Jessup, 2010; Lovell, 2008; Smith and Kern, 2009; Szarka, 2004; Welch et al., 2018) with Welch et al. (2018) also using workshops. Separately, Leifeld and Haunss (2012) used quantitative social network approaches.

This analysis is based on a wider exploratory study which considered the issue of incumbency in the GB heat sector. This study was based around in-depth interviews which took places across 2017 and a review of grey literature, both of which have value for CDA and DC approaches.

Interviewees included individuals from the private sector, third sector and public sector, all with an interest in and knowledge of heat decarbonisation and the role of industry. A previous review highlighted key business sectors and large businesses operating in the GB heat sector (Lowes et al., 2018a). This mapping allowed targeting of interviews in order to gain representation across the GB heat sector. Those involved in policy development were also identified from the research team's existing network. A snowballing approach was used to reach further relevant interviewees. This interview data was coded into themes using NVivo software.

This paper takes an ex-post approach to analysis of the DC based on findings from the wider research. Based on interview data from 30 interviews⁷ and with the inclusion of relevant grey data, the results describe the DC storyline, the associated members and the practices used to attempt to give power to the discourse. Following the emergence of a storyline, we critically asses the storyline based around the best available data and analysis on heat decarbonisation in the UK.

Hajer (2006) also suggests that DCs can be considered to be dominant if central actors are forced to accept the new discourse and the discourse is reflected in relevant political institutions. While we don't focus on the dominance of the incumbent storyline, we pay close attention to the requirement for further careful analysis, and the policy implications of our findings.

All interviews have been anonymised, however where quotations are used, the type of interviewee may be included, if relevant. Annex 1 contains a list of interviewees detailing the type of interviewee and date of interview.

4. The discourse coalition

In building on Hajer's approaches and like Welch et al. (2018), this section considers the DC and its central storyline taking account of the key incumbent actors, the storyline being presented, and the practices used to reproduce the storyline.

4.1. Actors and the coalition

Interviews and grey literature analysis highlighted a group of actors promoting a heat future in which the combustion of gas forms the central tenet of a future low carbon heat system. This coalition was formed primarily of incumbent businesses with an interest in the GB gas sector. Key actors which emerged were gas boiler manufacturers, gas distribution networks and associated lobby groups.

A number of interviewees suggested the idea of converting the gas grid to 100 % hydrogen to replace 100 % of heat demand in buildings, was being promoted primarily by the gas network owners (four interviewees). The role of the gas appliance (boiler) manufactures in promoting the low carbon gas storyline was also suggested (four interviewees). There was also a recognition among a number of interviewees that the idea of converting the gas grid to 100 % hydrogen had emerged fairly recently (four interviewees).

Interviews also highlighted that the off-grid liquefied petroleum gas (LPG) industry was another group of actors promoting 'green gas' (2 interviews) with one heat pump interest suggesting that this was to protect LPG from being replaced by electric heat pumps in rural areas. A 2017 publication from the UKLPG trade body supports 'bio-propane' as an off gas grid green gas solution (UK LPG, 2017).

The presence of a self-identifying coalition which refers to itself as the 'decarbonised gas alliance' also emerged from interviews (three interviews). A presentation given to The UK Department for Business, Energy and Industrial Strategy (BEIS) by the alliance which was shared by an interviewee showed that this alliance includes specific hydrogen interests, upstream gas producers, consultants and researchers as well as the previously mentioned gas appliance manufacturers and networks incumbents. The categories and their members as shown in this presentation (available on request) are listed below:

- Hydrogen, including transport and end use: ITM Power, Worcester Bosch Group, Kiwa, Providence Policy, ULEMco.
- CCUS: Carbon Capture and Storage Association, Cambridge Carbon Capture, Pale Blue Dot, Summit Power.
- Gas networks, energy trade associations: Northern Gas Networks, National Grid, SGN, Wales and West Utilities, Energy Networks

⁷ From a wider set of 51 interviews. The interviews not included did not contain data relevant to the research questions.

Association, EUA, Energy UK, UKOOG, East of England Energy Group.

- Industrial, scientific: Johnson Matthey, BOC, Ineos, Chemical Industries Association, Peel.
- Engineering, standards, consulting: Arup, Institution of Mechanical Engineers, IGEM, DNV-GL.
- Local government: Tees Valley Combined Authority.
- Energy companies: Total, Shell, Spirit Energy.
- Academia, research: GERG The European Gas Research Group, Sustainable Gas Institute, University of Strathclyde Oil and Gas Institute, Energy Research Accelerator, Birmingham Energy Institute, The Tony Davies High Voltage Laboratory, University of Chester, Powerful Women.

This relatively organised coalition previously had no publicly available profile or website. In the presentation to BEIS, in order to increase policy support to levels perceived to be being given to the renewable energy sector, the group explain: '*We believe that the ambition for the decarbonisation of the gas system should match the ambition for the continuing development of renewables*' (Decarbonised Gas Alliance, 2018).

The organisation now has a website which explains that it represents companies including Shell, Total, Equinor and the bodies which represent UK gas network owners and gas heating appliance manufacturers and is run by company DNV GL (Decarbonised Gas Alliance, 2020). The website also highlights the organisation's interests in biogas and hydrogen.

It is apparent that a clear coalition, part of which appears to have formalised, promoting the green gas storyline, exists. This coalition includes not just incumbent business interests but wider interests who appear to have coalesced around the green gas story line. This reflects Szarka's (2004) and Bulkeley's (2000) observations that DC members may not necessarily share world views but become a coalition around a storyline.

Two interviewees explained that the shale gas lobby group 'UK Onshore Oil and Gas' originally convened the Decarbonised Gas Alliance but the group is now run by consultants DNV GL. Following a request, DNV GL did not provide any further information on the group (corresponding author, personal communication, 5/2/18).

4.2. The storyline

The 'green gas' storyline is fundamentally based on the principle that the gas currently used for heating in GB can be replaced with a lower carbon gas. Under this storyline, in a decarbonised world, the gas grid in GB is maintained and gas connected households continue to combust gas in boilers in a similar way as they do today.

Interviews highlighted the potential for a number of different low carbon gas options:

- Converting the gas grid to run on 100 % hydrogen (seven interviewees);
- Blending a percentage of hydrogen into the fossil gas grid (two interviewees);
- Using synthetic natural gas produced from solid fuels (two interviewees);
- Replace fossil gas and liquefied petroleum gas with gas from bioenergy sources (biomethane or bio liquefied petroleum gas) (four interviewees).

A key element of the storyline is that the green gas option represents continuity for consumers who would not need new types of appliances whereas for non-gas technology options, new appliances such as heat pumps would be needed (four interviewees).

For consumers, green gas was suggested by a firm involved in manufacturing gas boilers to represent 'business-as-usual, because they can relate to a boiler, they can understand it, they know how it works'. One interviewee from the third sector explained that the use of this idea of consumer continuity allowed incumbents to 'make their case [for the green gas storyline] via consumer interest'.

This idea of minimising disruption is also visible in industry produced grey literature. For example industry analysis into converting the Leeds gas system to run on hydrogen explains:

'The process will involve minimal disruption for the customer (domestic or commercial) and require no large scale modifications to their property' (Northern Gas Networks et al., 2016, p1).

According to a GB boiler manufacturer's lobbying strategy shared with the author, green gas is:

'the most cost effective and pragmatic way of decarbonising the UK's heat supply, whilst simultaneously delivering a solution that will be most acceptable and cause the least disruption to consumers'.

In a consultation response, The Energy Networks Association (ENA) which represents gas and electricity network owners explains that 'injecting green gas such as hydrogen into the grid, offers significant cost savings against alternative low carbon heating sources. It is also shown to be the most practical scenario in terms of technical feasibility and, importantly, acceptance from customers and society' (ENA, 2017).

Overall, the low carbon gas storyline maintains the gas system and is framed around representing continuity and minimal disruption for consumers.

4.3. Practices

The practices of the green gas DC highlighted from interview data appear primarily related to influencing energy policy and future heat system visions. These key practices identified were associated with engagement with policy makers and the publication of reports supportive of green gas pathways.

One civil servant explained:

'It's [the gas industry] a huge industry; they are very powerful companies and all powerful companies particularly in a highly regulated market have a huge government lobbying and influencing role to protect their position for the future. I mean, clearly they are having a big influence the whole time'.

4.3.1. Engagement with policy makers

Interviewees, including those from incumbent firms, described engagement with Government by gas and oil boiler manufacturers both directly (three interviewees) and also through trade associations (six interviewees) with the goal of promoting green gas options and protecting interests. A large number of interviewees also referred to lobbying and regulatory pressure associated with the gas network companies (six interviewees). This lobbying was suggested to be linked to concerns over the potential for decarbonisation to reduce the requirement for gas infrastructure (6) with one interviewee explaining 'they've got terrified that the gas networks are going to become redundant' as a result of heat electrification.

One interviewee from a large gas and oil boiler manufacturer explained: 'we do try hard to influence departments, obviously BEIS now and ministers' and provided the interviewer with a document described as the company's 'Heat lobbying strategy'. This strategy proposed focusing on converting the gas grid to run on low carbon gas and using bio-energy for homes not connected to the gas grid.

The gas networks were described by an energy researcher from an independent research organisation as 'lobbying, very, very, very hard' on hydrogen with another interviewee from a large energy supplier explaining 'I mean it's often the gas networks who are leading this charge for quite obvious reasons'. One interviewee from a gas network company explained that they actively lobby Ofgem, the GB network regulator, who controls investment in the gas grid, to show a need for continued investment. In their own words:

'for them [Ofgem] to sanction kind of eight years' worth of investment in the gas distribution network, and the gas transmission network, what we need to be doing is justifying to our independent regulator that actually that investment is going to deliver value for money, for users. So all the work that we're doing now, is building the story to show this is why you should invest in the gas networks, because that will actually be a valuable investment, on the path to 2050'.

Two interviewees from gas networks also explained that as well as trying to influence regulator Ofgem, they also attempt to influence central Government departments. One gas network interviewee explained:

'Our aim has got to be to influence the regulator to support the investments. But actually the policy makers are BEIS and DECC (The Department of Energy and Climate Change which has now been replace by BEIS) - was DECC, now BEIS - so what we're aiming to do is develop the objective, independent evidence; provide them with the understanding that actually you shouldn't be ruling out the gas network or selecting this pathway at the minute because we don't really know what the solution is'.

Details of civil service meetings with external organisations are not recorded however the Department for Business, Energy and Industrial Strategy does publish data on ministerial meetings. Although this data set is not complete, for the period from July 2017 to July 2018 this data includes three meetings attended by the 'Energy and Utilities Alliance' trade body (BEIS, 2018b).

These meetings were:

- A meeting on the 16th October 2016 along with other attendees to discuss 'domestic heat and energy efficiency';
- A meeting alongside other attendees on 11th January 2018 to discuss 'oil and gas'
- A meeting alongside other attendees on 7th March 2018 to 'discuss the hydrogen economy'

The EUA represents a number of gas interests including some gas networks and gas boiler manufacturers. It is seen to be a particularly active pro-hydrogen lobby group according to an interviewee from a heating appliance manufacturer which produced both fossil fuel and low carbon heating systems.

The 11th January meeting was also attended by the ENA which represents UK gas networks owners⁸ who appear to be part of the green gas DC. The ENA has recently announced a new cross-network initiative 'Gas Goes Green' to promote and support the decarbonisation of the gas, suggesting their mission can deliver '*net zero in the most cost effective and least disruptive way possible*' (ENA, 2020).

One interviewee from an energy consultancy firm explained that the EUA trade body had employed an ex-Member of Parliament to promote the industry: 'the guy who was the MP, I can't remember his name' referring to the organisation's chief executive. The concept of the 'revolving door' whereby individuals move between Government and lobbying firms has been discussed elsewhere and is a phenomenon which allows individuals to take contacts and expertise from Government to other sectors to support influencing activities (Vidal et al., 2012).

The EUA released a manifesto in 2017 in which they explained: 'the UK has the world's leading gas grid infrastructure in place, directly supplying the energy to heat 85 per cent of UK homes. It would be a travesty not to use this existing infrastructure as part of the solution to the trilemma, and "green" gas could be the key.' (EUA, 2017). The EUA continues, on a video on their website that: 'Using a range of green gases, we can offer a cost effective way of meeting our 2050 carbon reduction targets' (EUA, 2019).

The website of the EUA makes clear that it does respond to Government consultations and, the twitter account of the Heating and Hot Water Industry Council, part of the EUA explains in a tweet on 11th June 2018 '*Green gas is the best means to #decarbonise #heat and tackle the #energy trilemma. Read why in our response to @beisgovuk' call for evidence 'a future framework for heat in buildings' which can be found here -https://goo.gl/vCaZyc'.* While the consultation focused on homes off the gas grid, the 'Heating and Hot Water

⁸ ENA also represents electricity network owners.

Industry Council' explain that they responded to BEIS, 'informing them that decarbonising the gas grid, using green gas offers the most pragmatic, practical and cost effective solution to the energy trilemma' (HHIC, 2018).

Our analysis has also highlighted the response to consultations by the previously identified Decarbonised Gas Alliance. Specifically, the DGA responded to a call for evidence from the UK's CCC on reaching a zero carbon economy. In their response, the DGA promoted the role of hydrogen for buildings currently connected to the gas-grid and biogas (bio-LPG) for buildings in off-gas grid areas (Committee on Climate Change, 2019c).

It has also emerged that a network of gas interests which includes GB's four regulated gas distribution networks and the EUA trade association have worked together to develop an 'All Party Parliamentary Group on hydrogen' led by public affairs consultancy Connect PA (Connect Public Affairs, 2018). All Party Parliamentary Groups are not official parliamentary bodies but provide a forum for engagement on particular issues and are open to all members of parliament and peers (Parliament, 2018).

4.3.2. Funding reports and analysis

An important element of practices to promote the green gas storyline appears to be through the production of reports which contain visions of the future heat system where an important role of gas is emphasised.

Five such reports were discovered. These were:

- '2050 Energy Scenarios: The UK Gas Networks role in a 2050 whole energy system', carried out by consultants KPMG and funded by the ENA (KPMG, 2016). This report suggested that an 'evolution of gas' scenario which saw no growth in electric heating but with much fossil gas supply replace by hydrogen and biomethane had lower practical obstacles and would be significantly cheaper that technology options which contained a greater proportion of heat electrification. Funders the ENA explain the report shows that 'Making use of the UK's gas network infrastructure offers a practical and affordable solution to the future challenge of heat decarbonisation' (ENA, 2016);
- 'The Green Gas Book' published by the Parliamentary Labour Party but funded by the ENA with contributions from various gas industry interests. This book suggested that 'green gas has the potential to help us pragmatically solve the immense energy and climate challenges we face. It deserves our full support' (The Parliamentary Labour Party Energy and Climate Change Committee, 2016, p8);
- 'Too hot to handle' produced by think tank Policy Exchange and funded by the EUA and Calor Gas which called for a 'more balanced set of priorities and technologies-incorporating substantial improvements in energy efficiency, more efficient gas appliances, greener forms of gas...' (Policy Exchange, 2016, p7);
- 'Next steps for the gas grid' produced by think tank Carbon Connect and funded by the Institute of Gas Engineers and Managers which concluded that 'the gas grid could play a vital role in transitioning to a low carbon energy system through the widespread use of low carbon gas' (Carbon Connect, 2017, p9);
- 'Energy from gas: taking a whole system perspective' produced by the Institute of Mechanical Engineers with contributions from The Institute of Gas Engineers and Managers, the ENA and gas network owner Wales & West Utilities which called for investment and further research into hydrogen (Institute of Mechanical Engineers, 2018);

5. Critical analysis and the uncertainty of 'green gas'

This section critiques the green gas storyline being presented by the incumbent led coalition. This critical approach is based on available evidence around low carbon gases as well as a recognised need for rapid decarbonisation and an analytical position based on existing evidence that electrification is important for heat decarbonisation. As highlighted in section 1.1, much analysis and indeed meta-analysis has suggested that a significant degree of heat electrification is needed for decarbonisation in the UK. While we don't suggest that electrification ever had hegemonic status in the UK heat decarbonisation discourse, clearly it appears to be an important element.

The socio-political relevance of this critique is that our prior analysis has shown the development of an organised and incumbent led coalition promoting a solely low carbon gas pathway in spite of the recognised need for significant electrification. The discourse coalition may be over-selling the potential for 'green' gas while at the same time sitting in a privileged discursive position, linked to incumbent capacity and institutional integration.

The following sections primarily critique the technological pathways promoted by the discourse coalition. Attention is also paid to ideas of continuity for consumers, reduced disruption and lower costs, all elements which have been suggested to be associated with low carbon gas options.

5.1. Converting the gas grid to 100 % hydrogen

Converting the UK's gas grid to hydrogen was the key approach suggested by the DC to achieve significant levels of carbon reduction in heating for areas with gas grid infrastructure. This approach generally sees hydrogen being produced from fossil gas using reforming processes with the carbon content of the gas captured and stored with carbon capture and storage (CCS) technologies (HM Government, 2017; Northern Gas Networks et al., 2016). Hydrogen is then generally expected to be burnt in suitable boilers which operate in a similar way to gas boilers but new appliances would be required; as would some sort of programme which converts certain geographical areas from fossil gas to hydrogen at different times (Northern Gas Networks et al., 2016).

Despite the promotion of this option, a gas system has never been converted to low carbon hydrogen anywhere in the world and there are major uncertainties around potential carbon reduction, costs and technical feasibility. The uncertainties are expanded in the

following sub-sections.

5.1.1. Carbon reduction potential

In order to be a suitable option for heat decarbonisation, levels of GHG emissions from a hydrogen system must be minimal. This is a fundamental point, yet hydrogen production may have significant associated emissions.

The Leeds Citygate research project, delivered by a UK gas network, suggested that the conversion of Leeds to hydrogen would result in carbon reductions of only 60 % compared to fossil gas due to emissions from the hydrogen production and carbon capture process (Northern Gas Networks et al., 2016). Peer reviewed analysis investigating the carbon credentials of hydrogen suggests that the lifecycle GHG emissions of hydrogen produced using fossil gas with CCS for heating can vary significantly from 23 to 150 gCO₂eq/kWh; fossil gas lifecycle emissions for heating are around 230–318 gCO2eq/kWh (Balcombe et al., 2018).

The potential carbon intensity of hydrogen produced from CCS is uncertain. Based on the previous figures, the most optimistic GHG reduction potential of hydrogen compared to fossil gas (from 318 gCO2eq/kW reduced to 23gCO2eq/kW) is 93 % though the most pessimistic is a reduction of 35 % (from 230 gCO2eq/kW reduced to 150gCO2eq/kW). If extremely low carbon hydrogen cannot be produced, it is not clear what role it can have in a net-zero world. Even if a 93 % reduction in the emissions from fossil gas could be achieved with hydrogen, the remaining emissions would need to somehow be offset.

While theoretically, near zero levels of carbon emissions could be achieved by producing hydrogen from renewable and low carbon electricity using electrolysis, techno-economic analysis has suggested this would cost significantly more than a more direct electrification route where electricity is used directly for heat in buildings (Strbac et al., 2018).

A fundamental question remains over whether hydrogen, produced from methane alongside CCS, can reduce emissions in line with carbon reduction goals. This question may require practical trials in order to be answered.

5.1.2. Cost uncertainty

The discourse coalition suggested that converting the gas grid to hydrogen would be cheaper than an electrification based approach. The lack of practical hydrogen conversion projects and the scale of the potential conversion programme means that all costs are estimated and uncertain. Two pieces of analysis by Government advisors have investigated potential costs of hydrogen as a heat decarbonisation option:

- Analysis by the National Infrastructure Commission into space and hot water heating suggested that there were very large uncertainties associated with the costs to decarbonise heat although converting the gas grid to hydrogen from fossil gas with CCS could be cheaper than the alternative of electrification (Element Energy and E4tech, 2018). However, following sensitivity analysis, the authors explain '*no pathway can definitively be ruled the lowest cost option*' (Element Energy and E4tech, 2018, p86). Specifically on the uncertainties of hydrogen, the report goes on: '*Cost-effective hydrogen heating is highly likely to be reliant on carbon capture and storage (CCS), which is also as yet unproven, and carries substantial cost uncertainty*' (p9). Within this analysis, hydrogen was only expected to reduce emissions from heat by around 75 % and so while hydrogen may appear potentially cheaper, carbon reduction potential is not seen to be as high. Because this analysis was carried out before the introduction of the UK net-zero goal (when the target was an 80 % reduction in GHG emissions by 2050 on 1990 levels) and because of the residual emissions from CCS, this analysis may now only have limited value because it doesn't model costs for a very low carbon hydrogen system.
- Analysis by researchers at Imperial College London on behalf of the Committee on Climate Change into space and hot water heating suggested that, electrification or hybridisation (incorporating electrification) approaches appeared cheaper than hydrogen conversion scenarios for deeper decarbonisation although costs were similar at lower levels of carbon reduction. For a zero emission scenario, full electrification was modelled to cost £92 billion per annum compared to £121.7 billion per year for a conversion which sees the current gas based heat demand converted to hydrogen⁸ (Strbac et al., 2018).

Despite apparent cost uncertainties, gas network owner Northern Gas Networks explain in a promotional video: '*The H21 Leeds Citygate project has established that converting a city like Leeds is both technically possible and financially viable*' and that low carbon hydrogen can be produced at an '*extremely modest cost*' (Northern Gas Networks, 2018). This member of the green gas DC may be overlooking some of the apparent uncertainty around hydrogen.

The costs for any form of major transition are likely to be uncertain because of the scale of change and wider landscape level impacts such as technology cost reductions. Global falls in the cost of renewable energy and batteries mean that decarbonisation via electrification may now be cost less than was previously assumed (Committee on Climate Change, 2019a). The promotion of hydrogen as an obviously cost-effective option for low carbon heat is not reflected in current analysis and remains deeply uncertain.

5.1.3. Technical uncertainty

A lack of real world experience means that technical uncertainty around low carbon hydrogen production and use exists. A key uncertainty is the technical feasibility of large scale carbon capture and storage which is seen as deeply uncertain globally (Middleton and Yaw, 2018) and a key uncertainty for the UK's future energy system (Ketsopoulou et al., 2019).

There is also uncertainty around the safety of hydrogen, the suitability of pipework in people's homes and the development of appliances to burn hydrogen (Lowes and Woodman, 2020). There are also potential uncertainties around the sourcing of fossil gas feedstocks for hydrogen production with the requirement for fossil gas demand to increase due to hydrogen production losses/leakage. The industry led Leeds Citygate project suggest that 47 % more fossil gas would be needed to provide the same amount of

energy from hydrogen as is currently supplied by fossil gas (Northern Gas Networks et al., 2016), a potential issue recognised in a review of UK gas security (Bradshaw, 2018).

As a novel approach, the idea of hydrogen conversion is fraught with technical uncertainty. While research is underway to reduce this uncertainty (including Hy4Heat, 2020 which is looking at the viability and safety of hydrogen boilers), if hydrogen conversion isn't technically possible, or comes up against unexpected hurdles, relying on it as a strategy represents a high risk approach. Further still, despite suggestions from the DC that low carbon gas may present only limited consumer disruption, the apparent requirement for geographically based conversions, new appliances and issues with internal pipework may mean that the potential disruption to householders is significant.

Despite this deep technical uncertainty, hydrogen is being presented by the discourse coalition as a workable option with suggestions that other technologies based around electrification shouldn't be deployed. With no practical trials of fully hydrogen areas, we suggest that relying on this technology and delaying the deployment of known technologies represents a high risk approach leading to the potential delay of the deployment of technologies with much more better known cost, performance and immediate decarbonisation potential (Rosenow and Lowes, 2020).

5.1.4. Blending hydrogen with fossil gas

Our DC analysis highlighted the promotion of the potential for blending hydrogen into the fossil gas grid. However, despite being promoted as a widely valuable approach, blending is naturally limited to only partial (relatively minimal) decarbonisation.

Firstly there are limits on how much hydrogen can be blended with fossil gas due to the behaviour of gases in appliances when combusted. Research in the United States has suggested an upper limit of around 25 % hydrogen by volume may be possible for blending situations (Melaina et al., 2013). This upper limit is due to the changing energy content and combustion properties of gas as hydrogen is added to methane; these include wobbe index and flame velocity (de Vries et al., 2017d). A current UK trial is attempting a blend of up to 20 % by volume in a discreet UK gas network (HyDeploy, 2019).

The potential upper bound on the level of hydrogen blending implies the fossil gas would be required in concentrations of at least 75 % limiting the potential for decarbonisation. Compounding this issue is the fact that hydrogen has an energy density around 3 times lower than methane (de Vries et al., 2017d). This therefore means that if 25 % zero carbon hydrogen was blended with fossil gas, this would only reduce emissions from gas by around 8%.

It is therefore very unlikely that hydrogen/methane blends could play any role in a net zero UK energy economy. Hydrogen could potentially be blended with low carbon methane but there are also uncertainties over the availability of biogases (see section 5.2).

There appears to be a suggestion by some incumbents, such as gas network owner Northern Gas Networks that blending could *'pave the way for a clean, low carbon gas grid'* (Northern Gas Networks, 2016) and blending could deploy some hydrogen production technologies and create wider market learning. While some learning may be possible, because blending utilises existing infrastructure and appliances, it seems unlikely to provide learning around the network and domestic requirements for full hydrogen conversion. Blending therefore has only limited carbon reduction potential and it is not clear that wider benefits will accrue around its deployment.

5.2. Biogas

The discourse coalition identified has also been promoting biogenically sourced gases as a means to decarbonise heating. Biomethane is one form of low carbon gas which has already seen growth across Europe and the UK. Biomethane is biogas, produced from certain types of biomass, most often via anaerobic digestion (AD), which is treated and injected into the gas grid where it functions exactly like fossil gas (Speirs et al., 2017). As of September 2019, there were 94 operating biomethane injection sites which inject treated biogas into the UK's gas network (BEIS, 2018c) and there are over 500 plants operating across Europe (European Biogas Association, 2018).

While biomethane is a technology which is currently operational, the type of bioenergy feedstocks used to produce biomethane can have a significant impact on the carbon intensity of the resulting biomethane; for the lowest carbon biomethane, waste feedstocks are required rather than purpose grown energy crops (Adams et al., 2015).

The national availability of suitable biomass resource for large-scale UK production of biomethane from AD is likely to be limited. Overall the CCC suggest that sustainable, UK produced biomass could meet between 5% and 10 % of UK energy demand by 2050, however within the buildings sector, bioenergy looks likely to only provide a niche role in a decarbonised system (Committee on Climate Change, 2018b).

Analysis of a net-zero UK energy system suggested that sustainable biomethane produced from AD, could meet a maximum of 5% of current building heat demand the optimal use of this resource was for peaking heat demand (for hybrids) or other niche roles such so in local heat networks (Committee on Climate Change, 2019d). It is also possible that biomethane could be of value in other sectors such as freight transport or industry where high temperatures are needed, further reducing the resource availability for domestic heat applications.

Another challenge for the carbon intensity of biomethane injection is propanation. In order to ensure the calorific value of injected biomethane is within the very fine tolerances set by gas network regulations, fossil propane may also be injected (Ofgem, 2018). Propanation increases the carbon intensity of biomethane operations although research is currently underway into potential approaches to remove this requirement (Ofgem, 2017).

The limited potential of biogas as a means to decarbonise heat has meant that as a technology, it hasn't featured in energy system cost analysis in the same way that hydrogen has. Fundamentally it doesn't appear to be able to be delivered at a scale of value to the

UK heat transformation. Speirs et al. (2017) also show that mean biomethane production costs currently appear to be significantly above hydrogen production costs (by around 50 %, based on steam methane reformation with CCS).

Despite the promotion of biogenic forms of gas, the potential for them to replace fossil gas appears at best limited and worst niche. This situation is exacerbated by the fact that only certain feedstocks for biogas, in particular waste, provide significant carbon savings. While costs are uncertain, biogas appears to also be a relatively expensive option.

5.3. Synthetic natural gas (SNG)

Analysis of the DC highlighted some promotion of this technology approach. There is only a very limited evidence base associated with the production and use of low carbon SNG for heating. This technique normally involves the gasification of solids followed by a chemical synthesis process.

One UK gas network owner Cadent, has proposed the production of synthetic gas from black bag waste suggesting that 'this technology would secure low carbon heat to millions of homes and businesses in a cost effective way' (Cadent, 2017, p1). Cadent is currently involved in a trial to produce SNG from 'refuse derived fuel' and inject in into the UK gas grid in Swindon although at the time of writing it is unclear that any gas has been injected.

The carbon intensity of synthetic natural gas depends on the feedstock use, plastics for example (made from oil) or the use of coal could result in a high carbon gas whereas some bio-energy sources could result in lower carbon intensities. According to Speirs et al. (2017), the first stage of the process, the production of syngas, can result in a gas with a variable carbon intensity depending on feedstock; however, the second stage to methanate the syngas into methane may undermine the benefits of the low carbon syngas by adding carbon into the process which will be released when the resulting methane is combusted.

The potential reliance on bio-energy feedstocks and unknown technical and cost performance means that like biogas, the role of synthetic natural gas as a low carbon heat option is limited. While a combination of biomass gasification and CCS could in theory result in a carbon sequestering fuel (i.e. negative emissions) (Speirs et al., 2017), the combination of technologies needed, the requirement for biomass and unknown costs make this approach deeply uncertain certainly not something that can be relied on either in the short term or at scale.

6. Discussion

Focusing on the issue of incumbency in the GB heat sector, and the emergence of a 'green gas' discourse in light of a perceived need for heat electrification, this article has set out to consider:

- 1 Can a DC can be identified and what practices are being used to promote the storyline?
- 2 How does the proposed storyline compare to relevant and recent analysis on UK heat decarbonisation pathways and the potential role for gas?
- 3 What are the transition and policy implications of the efforts of incumbents?

The existence of a DC is clear, and one element of the coalition self-identifies as an 'alliance'. A focus on the practices of the coalition, as suggested by Hajer (2006) has been valuable. The coalition appears to be formed of various actors ranging from some of the largest incumbents in the UK's heat sector currently involved in the gas regime to smaller actors such as consultants and universities.

The most active actors in the coalition appear to be those with business interests in producing gas, manufacturing gas appliances and owning and operating gas networks. However, as suggested by Lovell et al. (2009), it is apparent that actors have coalesced around a storyline which has value for them. This suggests that incumbent discourse coalitions can draw in actors, such as research organisations, who may reproduce storylines potentially enhancing the strength of the discourse further.

Our example clearly shows regime actors resisting change towards electrification highlighting ideas of regime resistance (Geels, 2014) and our study provides further analysis drawing together approaches to transitions and discourse coalitions as others have attempted (Späth and Rohracher, 2010).

This 'green gas' story is well aligned with the business interests of gas interests who are threatened by electrification, the competing heat decarbonisation pathway. The storyline appears to be promoted primarily through engagement with policy makers and through the related publication of analysis and reports.

The coalition's storyline is based around the idea of decarbonising heating by decarbonising the gas supply and maintaining gas infrastructure and the use of boilers in people's homes. The seductive element of this storyline is that it is framed as less costly and less disruptive to citizens compared to electrification. The promotion of a 'non-transformative' solution when required interventions may need to be much more transformative can be recognised as one of the 'discourses of delay' proposed by Lamb et al. (2020) where the discourse is based are the idea that 'disruptive change is not necessary'(p2).

However, our critical analysis has shown that there is significant uncertainty associated with the 'green' or low carbon gas pathway and disruptive and transformative change may be necessary. Of the technologies being promoted (biogenically sourced gas, hydrogen blending, synthetic natural gas, and full hydrogen conversion) only full hydrogen conversion appears to have the potential for significant decarbonisation potential. This is because both synthetic natural gas and biogas rely on bioenergy feedstocks which are limited in supply and expected to be needed elsewhere. Hydrogen blending is naturally limited to a small proportion of total energy throughput for technical reasons. Yet even though the conversion of gas infrastructure to 100 % hydrogen could possibly have the potential to deliver major GHG emission reductions, it will require large scale carbon capture and storage, new hydrogen production facilities and new appliances. The entire approach also has extremely uncertain costs and it has not been trialled at scale. An incumbent heavy coalition is overselling the green gas storyline and a fundamental question remains over whether the pathway is actually technically credible.

Our analysis has shown no evidence of incumbents attempting to actively block decarbonisation, but instead, approaches to resistance are based on reshaping the transition vision around the interests of the regime. Regime actors may be looking to capture the transition (Pel, 2015). It is possible that this apparent attempt to control change may be part of strategy to minimise overall change and maintain existing markets through the creation of uncertainty over options. This issue should be a key consideration for policy makers.

While a discourse coalition is clearly present and active, and the idea of hydrogen as a means to decarbonise heat has recently entered the GB heat decarbonisation policy discourse, it is not apparent that the green gas storyline dominates over other storylines. We previously suggested that despite being seen as important, electrification of heat doesn't appear to have shown hegemonic status. It is not possible at this stage to say that hydrogen is seen in policy discourse as the only means to decarbonise heating and indeed even UK Government documents recognise the two competing technological pathways (HM Government, 2017) and recent analysis highlights perceptions of uncertainty for policy makers around this issue (Lowes and Woodman, 2020).

This leads us on to some methodological reflections. We have found value in combining approaches and employing the lens of discourse coalitions, borrowed from ADA approaches with an overtly critical discourse analysis. Alone, the DC approach says little about the impact of the coalition and its potential impact on socio-technical system dynamics. However, when the discourse coalition and its storyline is critiqued using interdisciplinary analysis, useful insights on energy system transformations can be made. We therefore suggest that for those scholars interested in the acceleration of sustainable change, inter-disciplinary and explicitly critical discursive approaches may provide a useful method to unpick attempts by incumbents to (re)shape visions and policy discourses and consider policy impacts.

However, further micro-scale analysis of the policy changes associated with heat decarbonisation may be able to provide evidence of specific impacts of (incumbent) actors on policy change. A methodology based on triangulation of data sources may be of value (see Lowes et al., 2019 and Arts and Verschuren, 1999 for a potential methodology). Discursive approaches to investigate the influence of ideas in the policy process (such as discursive institutionalism, see Schmidt (2010)) may also have value in determining the embeddedness or hegemony of policy discourses within political institutions. Combinations of these methods and longitudinal analyses may be able to provide evidence of whether incumbent actors have successfully shaped policy discourses although of course resources available to researchers may limit the possibility of such complex and long term studies.

7. Conclusions and policy implications

The power of actors, and in particular incumbent actors on the dynamics of energy transitions, has featured frequently in considerations of socio-technical change. In particular, the power of actors to affect policy change has been suggested to be important. Yet specific analysis into how incumbents affect policy change is limited.

This article has described the existence and behaviour of a discourse coalition in GB formed primarily of gas industry incumbents, under threat from the decarbonisation of heating. This discourse coalition has presented a storyline to policy makers that an option which continues the use of the gas infrastructure and gas boilers but replaces fossil gas with some form of low carbon gas is the strategy which should be taken. This option is suggested by the DC to be technologically, economically and socially favourable compared to electrification.

While this analysis has shown the response of incumbents involved in gas heating to the threat of heat electrification, we note that some elements of the industry under study faces an existential threat and this may in part explain the forcefulness of their low carbon gas vision promotion. Not all incumbents will face such an existential threat and adaptation may be possible in other sectors leading to possibly more balanced responses.

Despite the promotion of the low carbon gas story line by incumbents, our critique shows that as well as being deeply technologically uncertain, the green gas storyline will not necessarily be cheaper or favourable. Indeed only one option for low carbon gas, hydrogen conversion, appears to be a serious contender compared to electrification and this technological approach is deeply uncertain. The green gas storyline is being oversold by incumbents. We make no prediction about optimal future energy systems in this paper but we note that electrification has be repeatedly been seen as an important low carbon heat option and can reduce greenhouse gas emissions immediately.

A key policy issue associated with the strong promotion of hydrogen is that it could have the effect of increasing uncertainty for policy makers therefore making any decisions difficult. The natural result of this could be a delay in policy development at a time when significant policy interventions are needed to drive rapid technological change in line with targets for net zero. Policy makers should be aware that attempts to increase uncertainty may be a strategy used by incumbents to delay the introduction of required decarbonisation policies.

While it comes as no huge surprise that low carbon gas is being strongly promoted by a gas industry under threat and that this industry engages with policy makers, a key concern for the transition to low carbon heating is the capacity that incumbents have to promote their storyline. It is not apparent from wider research around the issue of incumbency in the UK heat sector that niche actors have the capacity to compete with incumbents on policy and discursive issues (Lowes et al., 2018b).

This is a key structural issue associated with incumbency and while our analysis does not allow us to make specific recommendations on this issue we support the conclusion of (Lockwood et al., 2020) that greater awareness within Government of industry lobbying would be of value and the independent expertise should be valued. We also repeat calls for greater levels of transparency in UK policy making (Lowes et al., 2019).

In light of the UK's 2050 net zero target, heat needs to be decarbonised rapidly and any delay, caused by technological uncertainty, puts the UK target for GHG reduction at even further risk. While we recognise there is uncertainty, we suggest there is significantly less uncertainty over elements of electrification compared to hydrogen conversion. Low carbon heating appliances have been deployed at scale around the world and low carbon electricity is being generated. The same cannot be said for hydrogen boilers and low carbon hydrogen for heating.

To reduce uncertainty, independently verified, technical trials must rapidly attempt to reduce the uncertainties around hydrogen to see what, if any role, it can play in a sustainable heat system. Meanwhile, UK policy should focus on the deployment of known technologies which can decarbonise heating including heat networks, heat pumps and energy efficiency. Due to the uncertainties associated with hydrogen, in the short term, deployment of known low carbon heating technologies should be at a rate commensurate with the 2050 net-zero target with the expectation that low carbon gas including hydrogen may not prove viable at scale.

Declaration of Competing Interest

The authors report no declarations of interest.

Acknowledgements

We would like to thank all interviewees who took part in this research. We are particularly grateful to those interviewees who provided documentary evidence which was not publicly available. Thanks also to the anonymous reviewers whose thorough and thoughtful suggestions pushed this article towards a more critical and policy relevant position.

Funding

This research formed part of the programme of the UK Energy Research Centre and was supported by the Research Councils UK under the Engineering and Physical Sciences Research Council award EP/L024756/1.

Annex 1 List of interviewees by sector and date of interview

- Energy consultant with interest in heat 07/03/2017
- Director of heat interested trade body 05/04/2017
- External affairs manager of energy supplier 05/04/2017
- Technology manager from heating appliance manufacturer 06/04/2017
- Director at fossil and low carbon heat appliance manufacturer 07/04/2017
- Representative from gas network owner 10/04/2017
- CEO of heat and energy efficiency interested trade body 11/04/2017
- Energy consultant with an interest in heat 11/04/2017
- Policy analyst at energy regulator 12/04/2017
- Analyst with an interest in heat from consumer protection body 12/04/2017
- Analyst at non-governmental organisation interested in energy 13/04/2017
- Analyst at UK infrastructure advisor 13/04/2017
- Ex civil servant who worked on heat 19/04/2017
- Representative from gas network owner 24/04/2017
- CEO of heat interested trade body 05/05/2017
- Civil servant working on heat decarbonisation 05/05/2017
- Policy and communications manager for heating trade body 05/05/2017
- External affairs manager of heating appliance manufacturer 11/05/2017
- External affairs manager of boiler manufacturer 17/05/17
- Employees of government advisory body 18/05/2017
- Energy consultant with an interest in heat 30/05/2017
- Policy analyst at heat interested think tank 04/07/2017
- Policy and communications manager for heating trade body 05/07/2017
- Academic with research interest in heat 11/07/17
- Academic with research interest in heat 12/07/2017
- Director of low carbon appliance manufacturer 19/07/2017
- Policy analyst at heat interested think tank 25/07/2017
- Researcher at independent research organisation 27/07/2017
- Analyst at energy consultancy 29/10/2017
- External affairs manager and researcher from energy supplier 21/11/2017

References

- Adams, P.W.R., Mezzullo, W.G., McManus, M.C., 2015. Biomass sustainability criteria: greenhouse gas accounting issues for biogas and biomethane facilities. Energy Policy 87, 95–109. https://doi.org/10.1016/j.enpol.2015.08.031.
- Anshelm, J., Hansson, A., 2014. Battling Promethean dreams and Trojan horses: revealing the critical discourses of geoengineering. Energy Res. Soc. Sci. 2, 135–144. https://doi.org/10.1016/j.erss.2014.04.001.
- Arapostathis, S., Carlsson-Hyslop, A., Pearson, P.J.G., Thornton, J., Gradillas, M., Laczay, S., Wallis, S., 2013. Governing transitions: cases and insights from two periods in the history of the UK gas industry. Energy Policy 52, 25–44. https://doi.org/10.1016/j.enpol.2012.08.016.
- Arts, B., 2000. Political influence of NGOs on international environmental issues. In: Goverde, H., Cerny, P., Haugaard, M., Lentner, H. (Eds.), Power in Contemporary Politics. Sage, London, pp. 132–147.
- Arts, B., Verschuren, P., 1999. Assessing political influence in complex decision-making : an instrument based on triangulation. Int. Polit. Sci. Rev. 20, 411–424. https://doi.org/10.1177/0192512199204006.
- Avelino, F., 2009. Empowerment and the challenge of applying transition management to ongoing projects. Policy Sci. 42, 369–390. https://doi.org/10.1007/s11077-009-9102-6.
- Avelino, F., Rotmans, J., 2009. Power in transition: an interdisciplinary framework to study power in relation to structural change. Eur. J. Soc. Theory 12, 543–569. https://doi.org/10.1177/1368431009349830.
- Avelino, F., Rotmans, J., 2011. A dynamic conceptualization of power for sustainability research. J. Clean. Prod. 19, 796–804. https://doi.org/10.1016/j.jclepro.2010. 11.012.
- Balcombe, P., Speirs, J., Johnson, E., Martin, J., Brandon, N., Hawkes, A., 2018. The carbon credentials of hydrogen gas networks and supply chains. Renewable Sustainable Energy Rev. 91, 1077–1088. https://doi.org/10.1016/j.rser.2018.04.089.
- BEIS, 2018a. Clean Growth Transforming Heating Overview of Current Evidence [WWW Document]. URL. https://assets.publishing.service.gov.uk/government/ uploads/system/uploads/attachment_data/file/766109/decarbonising-heating.pdf.
- BEIS, 2018b. Ministerial Gifts, Hospitality, Travel and Meetings With External Organisations [WWW Document]. (accessed 8.23.18)URL. https://data.gov.uk/ dataset/d314604c-035b-42b2-99fb-b5a56fc4818d/ministerial-gifts-hospitality-travel-and-meetings-with-external-organisations.
- BEIS, 2018c. RHI Deployment Data [WWW Document]. URL (accessed 2.5.18). https://www.gov.uk/government/statistics/rhi-deployment-data-april-2017.
- Blommaert, J., Bulcaen, C., 2000. Critical discourse analysis. Annu. Rev. Anthr. 29, 447–466. https://doi.org/10.1146/annurev.anthro.29.1.447. 2000. Bradshaw, M., 2018. Future UK Gas Security : The Future Role of Gas [WWW Document]. URL. http://www.ukerc.ac.uk/asset/564F6D88-B4E6-40F9-87B72693DF4CCDE1/.
- Bulkeley, H., 2000. Discourse coalitions and the Australian climate change policy network. Environ. Plan. C Gov. Policy 18, 727–748. https://doi.org/10.1068/c3206r. Burnham, P., Gilland Luts, K., Grant, W., Layton-Henry, Z., 2008. Research Methods in Politics, 2nd ed. Palgrave Macmillan, Hampshire.
- Cadent, 2017. The Future of Gas [WWW Document]. URL. https://cadentgas.com/getattachment/About-us/The-future-role-of-gas/Doc-promo-Supply-of-renewable-gas/Cadent_Gas_Renewable_Gas.pdf.
- Caprotti, F., 2015. Golden sun, green economy: market security and the US/EU-China 'solar trade war. Asian Geogr. 32, 99–115. https://doi.org/10.1080/10225706. 2015.1057191.
- Carbon Connect, 2017. Next Steps for the Gas Grid [WWW Document]. URL. http://www.policyconnect.org.uk/cc/sites/site_cc/files/report/676/ fieldreportdownload/nextstepsforthegasgridweb.pdf.
- Castán Broto, V., 2015. Innovation territories and energy transitions: energy, water and modernity in Spain, 1939–1975. J. Environ. Policy Plan. 7200, 1–18. https://doi.org/10.1080/1523908X.2015.1075195.
- Chaliganti, R., Müller, U., 2016. Policy Discourses and Environmental Rationalities Underpinning India's Biofuel Programme. Environ. Policy Gov. 26, 16–28. https://doi.org/10.1002/eet.1697.
- Chaudry, M., Abeysekera, M., Hosseini, S.H.R., Jenkins, N., Wu, J., 2015. Uncertainties in decarbonising heat in the UK. Energy Policy 87, 623–640. https://doi.org/ 10.1016/j.enpol.2015.07.019.
- Clegg, S., Haugaard, M., 2009. Introduction. In: Clegg, S., Haugaard, M. (Eds.), The Sage Handbook of Power. Sage, London, pp. 1–24.
 Committee on Climate Change, 2010. The Fourth Carbon Budget: Reducing Emissions Through the 2020s [WWW Document]. 4th Budget/CCC-4th-Budget-Book_withhypers pdfl[B]. https://www.thecco.org.uk/archive/aws2/
- Committee on Climate Change, 2016. Next Steps for UK Heat Policy [WWW Document]. URL. https://www.theccc.org.uk/wp-content/uploads/2016/10/Next-stepsfor-UK-heat-policy-Committee-on-Climate-Change-October-2016.pdf.
- Committee on Climate Change, 2018a. Hydrogen in a Low-carbon Economy [WWW Document]. URL. https://www.theccc.org.uk/wp-content/uploads/2018/11/ Hydrogen-in-a-low-carbon-economy.pdf.
- Committee on Climate Change, 2018b. Biomass in a Low-carbon Economy [WWW Document]. URL. https://www.theccc.org.uk/publication/biomass-in-a-low-carbon-economy/.
- Committee on Climate Change, 2019a. Net Zero: the UK's Contribution to Stopping Global Warming [WWW Document]. URL. https://www.theccc.org.uk/wp-content/uploads/2019/05/Net-Zero-The-UKs-contribution-to-stopping-global-warming.pdf.
- Committee on Climate Change, 2019b. Reducing UK Emissions 2019 Progress Report to Parliament [WWW Document]. URL. https://www.theccc.org.uk/ publication/reducing-uk-emissions-2019-progress-report-to-parliament/.
- Committee on Climate Change, 2019c. DGA Response to CCC Consultation Into Building a Zero-carbon Economy [WWW Document]. URL. https://www.theccc.org.uk/wp-content/uploads/2019/04/Decarbonised-Gas-Alliance-response-to-Call-for-Evidence-2018.pdf.
- Committee on Climate Change, 2019d. Net Zero Technical Report [WWW Document]. URL. https://www.theccc.org.uk/wp-content/uploads/2019/05/Net-Zero-Technical-report-CCC.pdf.
- Connect Public Affairs, 2018. Gas Networks Back Parliamentary Group for Hydrogen Conversion [WWW Document]. URL. https://connectpa.co.uk/appg-hydrogen/gas-networks-back-parliamentary-group-hydrogen-conversion/.
- Cotton, M., Rattle, I., Van Alstine, J., 2014. Shale gas policy in the United Kingdom: an argumentative discourse analysis. Energy Policy 73, 427–438. https://doi.org/ 10.1016/j.enpol.2014.05.031.
- de Vries, H., Mokhov, A.V., Levinsky, H.B., 2017d. The impact of natural gas/hydrogen mixtures on the performance of end-use equipment: interchangeability analysis for domestic appliances. Appl. Energy 208, 1007–1019. https://doi.org/10.1016/j.apenergy.2017.09.049.
- Decarbonised Gas Alliance, 2018. Decarbonised Gas Alliance: Potential Sector Deal overview. London. available from author on request,
- Decarbonised Gas Alliance, 2020. Decarbonised Gas Alliance Contact [WWW Document]. URL (accessed 4.2.20). https://www.dgalliance.org/contact/.
- DECC, 2010. 2050 Pathways Analysis [WWW Document]. URL. https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/42562/216-2050-pathways-analysis-report.pdf.
- DECC, 2012. The Future of Heating: a Strategic Framework for Low Carbon Heat in the UK [WWW Document]. URL. https://www.gov.uk/government/uploads/ system/uploads/attachment_data/file/48574/4805-future-heating-strategic-framework.pdf.
- DECC, 2013. The Future of Heating : Meeting the Challenge [WWW Document]. URL. https://www.gov.uk/government/uploads/system/uploads/attachment_data/ file/190149/16_04-DECC-The Future of Heating Accessible-10.pdf.
- Earth Justice and The Sierra Club, 2020. Rhetoric vs. Reality: The Myth of "Renewable Natural Gas" for Building Decarbonization [WWW Document]. URL. https://earthjustice.org/sites/default/files/feature/2020/report-decarb/Report_Building-Decarbonization-2020.pdf.
- Element Energy and AEA, 2012. Decarbonising Heat in Buildings 2030-2050. Summary Report for the Committee on Climate Change. [WWW Document]. URL. . http://www.element-energy.co.uk/wordpress/wp-content/uploads/2012/04/Decarbonising-heat-in-buildings-summary-report-final_02.04.12.pdf.
- Element Energy, E4tech, 2018. Cost Analysis of Future Heat Infrastructure Options [WWW Document]. URL. https://www.nic.org.uk/wp-content/uploads/Element-Energy-and-E4techCost-analysis-of-future-heat-infrastructure-Final.pdf.
- ENA, 2016. KPMG Report Analyses Long-term Role of Gas Network in the Future of Heat [WWW Document]. URL (accessed 4.25.16). http://www.energynetworks.org/news/press-releases/2016/july/kpmg-report-analyses-long-term-role-of-gas-network-in-the-future-of-heat.html.
- ENA, 2017. Consultation Response: National Infrastructure Commission Call for Evidence [WWW Document]. URL Responses/Energy Networks Association National Infrastructure Assessment Call for Evidence 2017. pdf. http://www.energynetworks.org/assets/files/news/publications/Consultation.

- ENA, 2020. Gas Goes Green: Delivering the Pathway to Net Zero [WWW Document]. URL. https://www.energynetworks.org/assets/files/GGG_Launch_Doc_FINAL. pdf.
- EUA, 2017. EUA Manifesto 2017 [WWW Document]. URL. https://www.eua.org.uk/uploads/591995559CDBE.pdf.
- EUA, 2019. Green Gas [WWW Document]. (accessed 7.2.19)URL. https://eua.org.uk/green-gas/.
- European Biogas Association, 2018. GIE And EBA Publish the 'European Biomethane Map 2018' [WWW Document]. URL (accessed 8.24.18). http://european-biogas. eu/2018/02/14/gie-eba-publish-european-biomethane-map-2018/.
- Fairclough, N., 2010. Critical Discourse Analysis: the Critical Study of Language, 2nd ed. Routledge, Oxon.
- Fairclough, N., 2013. Critical discourse analysis and critical policy studies. #critedpol J. Crit. Educ. Policy Stud. Swart. Coll. 7, 177–197. https://doi.org/10.1080/ 19460171.2013.798239.
- Fuenfschilling, L., Truffer, B., 2014. The structuration of socio-technical regimes Conceptual foundations from institutional theory. Res. Policy 43, 772–791. https://doi.org/10.1016/j.respol.2013.10.010.
- Geels, F., 2010. Ontologies, socio-technical transitions (to sustainability), and the multi-level perspective. Res. Policy 39, 495–510. https://doi.org/10.1016/j.respol. 2010.01.022.
- Geels, F.W., 2011. The multi-level perspective on sustainability transitions: responses to seven criticisms. Environ. Innov. Soc. Transitions 1, 24–40. https://doi.org/10.1016/j.eist.2011.02.002.
- Geels, F., 2014. Regime resistance against low-carbon transitions: introducing politics and power into the multi-level perspective. Theory Cult. Soc. 31, 21–40. https://doi.org/10.1177/0263276414531627.
- Genus, A., 2014. Governing sustainability: a discourse-institutional approach. Sustainability 6, 283–305. https://doi.org/10.3390/su6010283.
- Hajer, M., 1993. Discourse coalitions and the institutionalization of practice: the case of acid rain in Britain. In: Fischer, F., Forester, J. (Eds.), The Argmentative Turn in Policy Analysis and Planning. Duke University Press, Durham and London, pp. 43–76.
- Hajer, M., 1995. Acid rain in Great Britain: environmental discourse and the hidden politics of institutional practice. In: Fischer, F., Black, M. (Eds.), Greening Environmental Policy: The Politics of a Sustainable Future. PCP, London, pp. 145–165.
- Hajer, M., 2002. Discourse analysis and the study of policy making. Eur. Political Sci. Rev. 2, 61-65. https://doi.org/10.1057/eps.2002.49.
- Hajer, M., 2006. Doing discourse analysis : coalitions, practices, meaning. In: Brink, M., Metze, T. (Eds.), Words Matter in Policy and Planning: Discourse Theory and Method in the Social Sciences. pp. 65–76 Utrecht: Koninklijk Nederlands Aardrijkskundig Genootschap.
- Haugaard, M., 2012. Rethinking the four dimensions of power: domination and empowerment. J. Polit. Power 5, 33-54. https://doi.org/10.1080/2158379X.2012. 660810.
- Hendriks, C.M., Grin, J., 2007. Contextualizing reflexive governance: the politics of dutch transitions to sustainability. J. Environ. Policy Plan. 9, 333–350. https://doi.org/10.1080/15239080701622790.
- Hewitt, S., 2009. Discourse Analysis and Public Policy Research [WWW Document]. URL. https://www.ncl.ac.uk/media/wwwnclacuk/centreforruraleconomy/files/ discussion-paper-24.pdf.
- HHIC, 2018. Green Gas at Heart of Future Heat Framework [WWW Document]. (accessed 10.28.19)URL. https://www.hhic.org.uk/news/green-gas-at-heart-offuture-heat-framework.
- HM Government, 2017. The Clean Growth Strategy: Leading the Way to a Low Carbon Future [WWW Document]. URL. https://www.gov.uk/government/uploads/ system/uploads/attachment_data/file/651916/BEIS_The_Clean_Growth_online_12.10.17.pdf.
- Hy4Heat, 2020. About Us [WWW Document]. URL(accessed 3.3.20). https://www.hy4heat.info/.
- HyDeploy, 2019. HyDeploy Website [WWW Document]. URL (accessed 7.4.19). https://hydeploy.co.uk/.
- Institute of Mechanical Engineers, 2018. Energy From Gas: Taking a Whole System Perspective [WWW Document]. URL. http://www.imeche.org/docs/defaultsource/1-oscar/reports-policy-statements-and-documents/imeche-energy-from-gas-report-final-may-2018.pdf?sfvrsn = 2.
- IPCC, 2018. Summary for policymakers. Global Warming of 1.5°C. An IPCC Special Report on the Impacts of Global Warming of 1.5°C Above Pre-industrial Levels and Related Global Greenhouse Gas Emission Pathways, in the Context of Strengthening the Global Response to [WWW Document]. URL. https://www.ipcc.ch/sr15/ chapter/spm/.
- Isoaho, K., Karhunmaa, K., 2019. A critical review of discursive approaches in energy transitions. Energy Policy 128, 930–942. https://doi.org/10.1016/j.enpol.2019. 01.043.
- Jessup, B., 2010. Plural and hybrid environmental values: a discourse analysis of the wind energy conflict in Australia and the United Kingdom. Env. Polit. 19, 21–44. https://doi.org/10.1080/09644010903396069.
- Johnstone, P., Stirling, A., Sovacool, B., 2017. Policy mixes for incumbency: exploring the destructive recreation of renewable energy, shale gas "fracking," and nuclear power in the United Kingdom. Energy Res. Soc. Sci. 33, 147–162. https://doi.org/10.1016/j.erss.2017.09.005.
- Kemp, R., Rotmans, J., Loorbach, D., 2007. Assessing the dutch energy transition policy: how does it deal with dilemmas of managing transitions? J. Environ. Res. Stat. Note Health Care Financ. Adm. Off. Policy Plan. Res. 9, 315–331. https://doi.org/10.1080/15239080701622816.
- Kenis, A., Bono, F., Mathijs, E., 2016. Unravelling the (post-)political in transition management: interrogating pathways towards sustainable change. J. Environ. Policy Plan. 7200, 1–17. https://doi.org/10.1080/1523908X.2016.1141672.
- Kern, F., 2011. Ideas, institutions, and interests: explaining policy divergence in fostering 'system innovations' towards sustainability. Environ. Plan. C Gov. Policy 29, 1117–1134. https://doi.org/10.1068/c1142.
- Kern, F., Smith, A., 2008. Restructuring energy systems for sustainability? Energy transition policy in the Netherlands. Energy Policy 36, 4093–4103. https://doi.org/ 10.1016/j.enpol.2008.06.018.
- Ketsopoulou, I., Taylor, P., Watson, J., Winskel, M., Kattiritzi, M., Lowes, R., Woodman, B., Poulter, H., Brand, C., Killip, G., Annable, J., Owen, A., Hannah, R., Gross, R., Lockwood, M., 2019. Disrupting the UK Energy System: Causes, Impacts and Policy Implications [WWW Document]. URL. http://www.ukerc.ac.uk/ publications/disrupting-uk-energy-system.html.
- Knobloch, F., Hanssen, S.V., Lam, A., Pollitt, H., Salas, P., Chewpreecha, U., Huijbregts, M.A.J., Mercure, J.-F., 2020. Net emission reductions from electric cars and heat pumps in 59 world regions over time. Nat. Sustain. 1–11. https://doi.org/10.1038/s41893-020-0488-7.
- KPMG, 2016. The UK Gas Networks Role in a 2050 Whole Energy System [WWW Document]. Future of Gas Main report plus appendices URL FINAL.pdf. http://www. energynetworks.org/assets/files/gas/tutures/KPMG.
- LA Times, 2019. California Ditched Coal. The Gas Company Is Worried It's Next [WWW Document]. URL (accessed 3.31.20). https://www.latimes.com/environment/ story/2019-10-22/southern-california-gas-climate-change.
- Lamb, W.F., Mattioli, G., Levi, S., Roberts, J.T., Capstick, S., Creutzig, F., Minx, J.C., Müller-Hansen, F., Culhane, T., Steinberger, J.K., 2020. Discourses of climate delay. Triplec Commun. Capital. Crit. Open Access J. A Glob. Sustain. Inf. Soc. 3, e17. https://doi.org/10.1017/sus.2020.13.

Lassen, I., 2016. Discourse trajectories in a nexus of genres. Discourse Stud. 18, 409-429. https://doi.org/10.1177/1461445616647880.

- Leifeld, P., Haunss, S., 2012. Political discourse networks and the conflict over software patents in Europe. Eur. J. Polit. Res. 51, 382–409. https://doi.org/10.1111/j. 1475-6765.2011.02003.x.
- Levidow, L., Papaioannou, T., 2013. State imaginaries of the public good: shaping UK innovation priorities for bioenergy. Environ. Sci. Policy 30, 36–49. https://doi.org/10.1016/j.envsci.2012.10.008.
- Lockwood, M., Mitchell, C., Hoggett, R., 2020. Incumbent lobbying as a barrier to forward-looking regulation: the case of demand-side response in the GB capacity market for electricity. Energy Policy 140, 111426. https://doi.org/10.1016/j.enpol.2020.111426.
- Love, J., Smith, A.Z.P., Watson, S., Oikonomou, E., Summerfield, A., Gleeson, C., Biddulph, P., Chiu, L.F., Wingfield, J., Martin, C., Stone, A., Lowe, R., 2017. The addition of heat pump electricity load profiles to GB electricity demand: evidence from a heat pump field trial. Appl. Energy 204, 332–342. https://doi.org/10. 1016/j.apenergy.2017.07.026.
- Lovell, H., 2008. Discourse and innovation journeys: the case of low energy housing in the UK. Technol. Anal. Strateg. Manag. 20, 613–632. https://doi.org/10.1080/09537320802292883.
- Lovell, H., Bulkeley, H., Owens, S., 2009. Converging agendas? Energy and climate change policies in the UK. Environ. Plan. C Gov. Policy 27, 90–109. https://doi.org/ 10.1068/c0797j.
- Lowes, R., Woodman, B., 2020. Disruptive and uncertain: policy makers' perceptions on UK heat decarbonisation. Energy Policy. https://doi.org/10.1016/j.enpol. 2020.111494.

- Lowes, R., Woodman, B., Fitch-roy, O., 2017. Defining Incumbency : Considering the UK Heat Sector [WWW Document]. URL. http://www.ukerc.ac.uk/asset/ 175A3A09-8AFF-43E7-898D3BE1846C07E9/
- Lowes, R., Woodman, B., Clark, M., 2018a. A Transformation to Sustainable Heating in the UK: Risks and Opportunities for UK Heat Sector Businesses [WWW Document]. URL. http://www.ukerc.ac.uk/publications/sustainable-heating-in-the-uk-risks-and-opportunities.html.
- Lowes, R., Woodman, B., Clark, M., 2018b. Incumbency in the UK Heat Sector and Implications for the Transformation Towards Low-carbon Heating [WWW Document]. URL http://www.ukerc.ac.uk/publications/incumbency-in-the-uk-heat-sector.html
- Lowes, R., Woodman, B., Fitch-Roy, O., 2019. Policy change, power and the development of Great Britain's renewable heat incentive. Energy Policy 131, 410-421. https://doi.org/10.1016/j.enpol.2019.04.041
- Lukes, S., 2005. Power: A Radical View, second edition. Palgrave Macmillan, Hampshire.
- Maclean, K., Sansom, R., Watson, T., Gross, R., 2016. Managing Heat System Decarbonisation: Comparing the Impacts and Costs of Transitions in Heat Infrastructure. Markard, J., 2018. The next phase of the energy transition and its implications for research and policy. Nat. Energy 3, 628-633. https://doi.org/10.1038/s41560-018-0171 - 7
- Melaina, M.W., Antonia, O., Penev, M., 2013. Blending Hydrogen Into Natural Gas Pipeline Networks : a Review of Key Issues Blending Hydrogen Into Natural Gas Pipeline Networks : a Review of Key Issues. https://doi.org/10.2172/1068610.
- Middleton, R.S., Yaw, S., 2018. The cost of getting CCS wrong: uncertainty, infrastructure design, and stranded CO2. Int. J. Greenh. Gas Control 70, 1–11. https://doi. org/10.1016/j.ijggc.2017.12.011
- worthern Gas Networks, 2016. £7 Million for Ground-breaking Green Heating Scheme at Keele University [WWW Document]. URL (accessed 10.28.19). https:// www.northerngasnetworks.co.uk/2016/11/30/7-million-for-ground-breaking-green-heating-scheme-at-keele-university/.
- Northern Gas Networks, 2018. H21 Leeds Citygate Film [WWW Document]. URL (accessed 8.24.18). https://www.northerngasnetworks.co.uk/ngn-you/the-future/ for-future-energy/
- Northern Gas Networks, Wales & West Utilities, Kiwa, Amec Foster Wheeler, 2016. Leeds City Gate H21 [WWW Document]. URL. http://www. northerngasnetworks.co.uk/wp-content/uploads/2016/07/H21-Report-Interactive-PDF-July-2016.pdf
- Ofgem, 2017. Project Summary: Future Billing Methodology [WWW Document]. URL. https://www.ofgem.gov.uk/ofgem-publications/107840.
- Ofgem, 2018. Guidance Volume Two: Ongoing Obligations and Payments [WWW Document]. URL. https://www.ofgem.gov.uk/system/files/docs/2018/10/ volume_2_for_publication_1.10.18_pdf_0.pdf.
- Parliament, 2018. Guide to the Rules on All-party Parliamentary Groups Published March 2015 and Slightly Revised May 2017 [WWW Document]. URL https:// www.parliament.uk/documents/pcfs/all-party-groups/guide-to-rules/guide.html (accessed 8.23.18).
- Pel, B., 2015. Trojan horses in transitions: a dialectical perspective on innovation 'capture. J. Environ. Policy Plan. 7200, 1–19. https://doi.org/10.1080/1523908X. 2015.1090903
- Policy Exchange, 2016. Too Hot to Handle [WWW Document]. hot to handle sept 16.pdfURL. http://www.policyexchange.org.uk/images/publications/too. Raven, R., Kern, F., Smith, A., Jacobsson, S., Verhees, B., 2016. The politics of innovation spaces for low-carbon energy: introduction to the special issue. Environ.
- Innov. Soc. Transitions 18, 101-110. https://doi.org/10.1016/j.eist.2015.06.008.
- Roberts, C., Geels, F.W., Lockwood, M., Newell, P., Schmitz, H., Turnheim, B., Jordan, A., 2018. The politics of accelerating low-carbon transitions: towards a new research agenda. Energy Res. Soc. Sci. 44, 304-311. https://doi.org/10.1016/j.erss.2018.06.001.
- Rosenow, J., Lowes, R., 2020. Heating Without the Hot Air: Principles for Smart Heat Electrification [WWW Document]. URL. https://www.raponline.org/wpcontent/uploads/2020/03/rap-rosenow-lowes-principles-heat-decarbonisation-march-2020.pdf.
- Schmid, S.D., 2004. Transformation discourse: nuclear risk as a strategic tool in late Soviet politics of expertise. Sci. Technol. Hum. Values 29, 353–376. https://doi. org/10.1177/0162243904264483.
- Schmidt, Va., 2010. Taking ideas and discourse seriously: explaining change through discursive institutionalism as the fourth 'new institutionalism. Eur. Polit. Sci. Rev. 2, 1. https://doi.org/10.1017/S175577390999021X.
- Smith, A., Kern, F., 2009. The transitions storyline in Dutch environmental policy. Env. Polit. 18, 78-98. https://doi.org/10.1080/09644010802624835.
- Smith, A., Stirling, A., Berkhout, F., 2005. The governance of sustainable socio-technical transitions. Res. Policy 34, 1491–1510. https://doi.org/10.1016/j.respol. 2005.07.005
- Smith, A., Voß, J.-P., Grin, J., 2010. Innovation studies and sustainability transitions: the allure of the multi-level perspective and its challenges. Res. Policy 39, 435-448. https://doi.org/10.1016/j.respol.2010.01.023.
- Späth, P., Rohracher, H., 2010. Energy regions': the transformative power of regional discourses on socio-technical futures. Res. Policy 39, 449-458. https://doi.org/ 10.1016/j.respol.2010.01.017.
- Speirs, J., Balcombe, P., Johnson, E., Martin, J., Brandon, N., Hawkes, A., 2017. A Greener Gas Grid: What Are the Options? [WWW Document]. URL. http://www. sustainablegasinstitute.org/a-greener-gas-grid/.
- Stenzel, T., Frenzel, A., 2008. Regulating technological change-the strategic reactions of utility companies towards subsidy policies in the German, Spanish and UK electricity markets. Energy Policy 36, 2645-2657. https://doi.org/10.1016/j.enpol.2008.03.007.
- Stirling, A., 2014. Transforming power: social science and the politics of energy choices. Energy Res. Soc. Sci. 1, 83–95. https://doi.org/10.1016/j.erss.2014.02.001. Strbac, G., Pudjianto, D., Sansom, R., Djapic, P., Ameli, H., Shah, N., Hawkes, A., 2018. Analysis of Alternative UK Heat Decarbonisation Pathways For the Committee
- on Climate Change [WWW Document]. URL. https://www.theccc.org.uk/wp-content/uploads/2018/06/Imperial-College-2018-Analysis-of-Alternative-UK-Heat-Decarbonisation-Pathways-Executive-Summary.pdf.
- Szarka, J., 2004. Wind power, discourse coalitions and climate change: breaking the stalemate? Eur. Environ. 14, 317–330. https://doi.org/10.1002/eet.367. The Parliamentary Labour Party Energy and Climate Change Committee, 2016. The Green Gas Book [WWW Document]. URL. https://alansenergyblog.files. wordpress.com/2016/07/13973-the-green-gas-book_96pp_v5.pdf
- Turnheim, B., Geels, F.W., 2019. Incumbent actors, guided search paths, and landmark projects in infra-system transitions: Re-thinking Strategic Niche Management with a case study of French tramway diffusion (1971-2016). Res. Policy 48, 1412-1428. https://doi.org/10.1016/j.respol.2019.02.002.
- UK LPG, 2017. Gas for Off-grid Britain [WWW Document]. URL. http://www.uklpg.org/uploads/DOC59F7391792193.pdf. UKERC, 2009. Pathways to a Low Carbon Economy: Energy Systems Modelling [WWW Document]. URL. http://www.ukerc.ac.uk/asset/6A6DE259-DAB0-4EE9-AA182A5F987A8927/
- Unruh, G., 2000. Understanding carbon lock-in. Energy Policy 28, 817-830. https://doi.org/10.1016/S0301-4215(00)00070-7.
- Van Dijk, T., 1993a. Principles of critical discourse analysis. Discourse Policy Soc. Sci. 4, 249-283. https://doi.org/10.1177/0957926593004002006.
- Van Dijk, T.A., 1993b. Principles of critical discourse analysis. Discourse Policy Soc. Sci. 4, 249–283. https://doi.org/10.1177/0957926593004002006.
- Van Dijk, T.A., 1995. The aims of critical discourse analysis. Japanes Discourse 1, 17–27. https://doi.org/10.1007/BF02092754.
- Van Dijk, Ta., 2001. Critical discourse analysis. The Handbook of Discourse Analysis. Blackwell, Oxord, pp. 352-371.
- Vidal, J.B.I., Draca, M., Fons-Rosen, C., 2012. Revolving door lobbyists. Am. Econ. Rev. 102, 3731–3748. https://doi.org/10.1257/aer.102.7.3731.
- Watson, S.D., Lomas, K.J., Buswell, R.A., 2019. Decarbonising domestic heating : What is the peak GB demand? Energy Policy 126, 533-544. https://doi.org/10.1016/ .enpol.2018.11.001.
- Welch, D., Swaffield, J., Evans, D., 2018. Who's responsible for food waste? Consumers, retailers and the food waste discourse coalition in the United Kingdom. J. Consum. Cult. https://doi.org/10.1177/1469540518773801.
- Winskel, M., Radcliffe, J., 2014. The rise of accelerated energy innovation and its implications for sustainable innovation studies: a UK perspective. Sci. Technol. Stud. 27.8-33.
- Winskel, M., 2016. From optimisation to diversity: changing scenarios of heating for buildings in the UK. In: Hawkey, D., Webb, J., Lovell, H., McCrone, D., Tingey, M., Winskel, M. (Eds.), Sustainable Urban Energy Policy: Heat and the City. Routeldge, Oxon, pp. p68-90.